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Status 
 Linac running in good condition for polarized proton and BLIP 

 

 -Last night,  replaced resistor in mod 5 modulator (3 HRS) 

 

 

 

 Maintenance (04/27/2015): < 6HRS) 

      - Collect water sample of tank 7 for analysis 

      - Install 5 kW solid state amplifier in Mod 6 

       - OPPIS laser maintenance (change crystal)  

           etc 
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